Author's response to reviews

Title: Cemented versus non-cemented hemiarthroplasty of the hip as a treatment for a displaced femoral neck fracture - design of a randomised controlled trial

Authors:

Anne JH Vochteloo (ajvochteloo@rdgg.nl)
DieuDonné Niesten (niesten@rdgg.nl)
Roeland Riedijk (riedijk@gmail.com)
Willard J Rijnberg (wrijnberg@alysis.nl)
Stefan BT Bolder (bolder_s@hotmail.com)
Sander Koëter (s.koeter@cwz.nl)
Keetie Kremers-van de Hei (k.kremers@cwz.nl)
Taco Gosens (tgosens@elisabeth.nl)
Peter Pilot (pilot@rdgg.nl)

Version: 5 Date: 18 May 2009

Author's response to reviews:

Title: Cemented versus non-cemented hemiarthroplasty of the hip as a treatment for a displaced femoral neck fracture – Design of a randomised controlled trial

Authors:
Anne J.H. Vochteloo (vochteloo@rdgg.nl)
DieuDonné Niesten (niesten@rdgg.nl)
Roeland Riedijk (riedijk@gmail.com)
Willard J. Rijnberg (wrijnberg@alysis.nl)
Stefan B.T. Bolder (bolder_s@hotmail.com)
Sander Koëter (s.koeter@cwz.nl)
Keetie Kremers-van de Hei (k.kremers@cwz.nl)
Taco Gosens (tgosens@elisabeth.nl)
Peter Pilot (pilot@rdgg.nl)

Date: 18 May 2009

Dear Editor,

Re: “Cemented versus non-cemented hemiarthroplasty of the hip as a treatment for a displaced femoral neck fracture – Design of a randomised controlled trial”

Thank you very much for accepting our manuscript.
We have carried through the major and minor revisions you suggested in the email.

On behalf of all authors;
Sincerely yours,

AJH Vochteloo MD